ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Meierrieks, Daniel; Stadelmann, David

Article — Published Version Is temperature adversely related to economic development? Evidence on the short-run and the long-run links from sub-national data

Energy Economics

Provided in Cooperation with: WZB Berlin Social Science Center

Suggested Citation: Meierrieks, Daniel; Stadelmann, David (2024) : Is temperature adversely related to economic development? Evidence on the short-run and the long-run links from sub-national data, Energy Economics, ISSN 1873-6181, Elsevier, Amsterdam, Vol. 136, pp. 1-18, https://doi.org/10.1016/j.eneco.2024.107758

This Version is available at: https://hdl.handle.net/10419/312587

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



http://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Contents lists available at ScienceDirect

Energy Economics

journal homepage: www.elsevier.com/locate/eneeco

Is temperature adversely related to economic development? Evidence on the short-run and the long-run links from sub-national data

Daniel Meierrieks^a, David Stadelmann^{b,*}

^a WZB Berlin Social Science Center, Reichpietschufer 50, 10785 Berlin, Germany
 ^b University of Bayreuth, Universitätsstraβe 30, 95447 Bayreuth, Germany

ARTICLE INFO

JEL classification: O44 O47 Q54 R11 Keywords: Regional temperature Regional per capita income Sub-national data Long-difference approach Threshold models Global warming

ABSTRACT

We examine the effect of rising temperatures on regional economic development, using annual sub-national data for over 1500 regions in 152 countries between 1990 and 2017. In a panel setting with region- and country-year fixed effects, we find no evidence of a homogeneous or heterogeneous effect of rising temperatures on economic development as measured by regional per capita income. Additionally, we find no non-linear relationship between temperature and economic development. We also employ a long-difference approach that is attuned to exploring the long-run relationship between rising temperatures and regional income. Results indicate that rising temperatures have a negative long-run impact on regional per capita income for a minority of regions located in countries with weak economic, legal and political institutions. Furthermore, these vulnerable regions located in development. The use of alternative regional per capita GDP data from 1950 onwards yields similar empirical results. Our findings suggest that negative disadvantaged countries. Thus, country-specific conditions may moderate regional economic vulnerability to future temperature increases due to global climate change.

1. Introduction

Over the past 200 years, per capita gross domestic product in most of the Western world has increased more than twenty-fivefold, while in much of the non-Western world it has grown more than tenfold (e.g., Deaton, 2013; Bolt and van Zanden, 2024). This economic development, in turn, was associated with improvements in human living standards and well-being, for example, in the form of higher life expectancy, reduced child mortality and lower malnutrition (e.g., Deaton, 2013; Weil, 2013). Promoting and maintaining future economic development is thus in humanity's vital interest. Indeed, a survey of experts by Christensen et al. (2018) predicts a global annual median 2010–2100 per capita income growth rate of 2.1%, suggesting that per capita incomes will, on average, increase more than fivefold over the remainder of the century.

However, there is substantial uncertainty associated with such estimates. A factor contributing to this uncertainty is global warming.¹ According to a recent report of the Intergovernmental Panel on Climate

Change (IPCC), the United Nations body for assessing the science related to climate change, during the 2011-2020 period average global surface temperatures were 1.09 °C higher compared to the 1850-1900 period (IPCC, 2021: SPM-5). What is more, the IPCC forecasts that average surface temperatures will be from 1.2 °C to 3.0 °C higher in the 2041-2060 period compared to the 1850-1900 period, with further increases being possible for the remainder of the twenty-first century (IPCC, 2021: SPM-18). Global warming, in turn, is expected to negatively affect human life. For example, it might curtail water availability and plant growth, thus jeopardizing food security, adversely affect health by contributing to the spread of diseases as well as fuel resource scarcity and subsequent political instability, especially in already more vulnerable parts of the world (IPCC, 2014). Consequently, global climate change is also expected to adversely affect global economic pathways. For instance, in a synthesis report, the IPCC, (2014: 16) projects that "[a] ggregate economic losses accelerate with increasing temperature [...] [and that] climate change impacts are projected to slow down economic growth [...]".

* Corresponding author.

https://doi.org/10.1016/j.eneco.2024.107758

Received 18 December 2023; Received in revised form 20 May 2024; Accepted 5 July 2024 Available online 10 July 2024

0140-9883/© 2024 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).





Energy Economics

E-mail addresses: daniel.meierrieks@wzb.eu (D. Meierrieks), david.stadelmann@uni-bayreuth.de (D. Stadelmann).

¹ Global climate change refers to the observed warming of the Earth's land and ocean surfaces. It is mainly due to anthropogenic greenhouse gas emissions such as carbon dioxide, methane and nitrous oxide (e.g., IPCC, 2014).

We contribute to the exploration of the adverse economic effects of global temperature rises by studying the relationship between rising temperatures and per capita levels of economic development with regional (i.e., sub-national) data for over 1500 regions in 152 countries between 1990 and 2017 and with an alternative economic dataset form 1950 onwards for robustness tests. The existing research on the nexus between temperature increases and economic outcomes at the sub-national level remains limited and inconclusive regarding statistical significance (e.g., Nordhaus, 2006; Dell et al., 2009; Zhao et al., 2018; Kalkuhl and Wenz, 2020; Greßer et al., 2021; Kahn et al., 2021; Kotz et al., 2024). We add to the empirical exploration of this nexus in three ways.

First, we use annual economic data from the *Global Data Lab Dataset* (Smits, 2016; Smits and Permanyer, 2019; Permanyer and Smits, 2020) which is available for many regions worldwide, especially within emerging and developing economies. As we measure regional economic activity by *regional per capita gross national income*, we can compare our results with the larger literature exploring the economic consequences of climate change in cross-country settings (e.g., Dell et al., 2012). In comparison to some earlier contributions on the regional temperature-income nexus (e.g., Greßer et al., 2021), we can fully exploit the panel structure of the data. This means that we can account for region-fixed effects as well as country-year-fixed effects that may correlate with warming and regional economic development (e.g., regional geographical conditions or national economic policy changes).

Second, we evaluate the long-run relationship between rising temperatures and regional economic development. A long-run perspective on the temperature-income relationship is warranted because climate change is commonly regarded as a cumulative and persistent phenomenon that may induce adaptation or intensification effects (e.g., Dell et al., 2014). Adaptation effects imply that warming may induce adaptive behavior (e.g., farmers may change to crops that are better adapted to changing climatic conditions), while intensification effects imply that economically damaging effects of climate change only materialize after longer time periods (e.g., as farmland may gradually desertify). Consequently, if adaptation effects prevail, we are likely to overestimate the link between regional warming and income when only considering the short run, while the prevalence of intensification effects means that we are likely to underestimate the same link when disregarding the long run. To uncover adaptation or intensification effects, we employ the long-difference approach of Dell et al. (2012, 2014) and Burke and Emerick (2016).

Third, we investigate potential heterogeneities in the temperatureincome relationship at the regional level by means of thresholdmodels. Previous cross-country research (e.g., Dell et al., 2012; Burke et al., 2015a) emphasizes that richer countries are less vulnerable to the adverse consequences of rising temperatures, as they potentially have the means (e.g., agricultural and health technology) available to adequately counter them. We add to this research by (1) considering the role of various economic and political institutions as potential moderators in the temperature-income relationship at the regional level, (2) examining the influence of moderators that have previously received no attention in the literature (e.g., differences between rural and urban areas), (3) studying heterogeneity in the regional temperature-income relationship in both the short and long run and (4) using a thresholdapproach following Hansen (1999) to empirically determine (rather than justifying in an ad-hoc manner) economic and political conditions under which the role of rising temperatures in regional economic development could become especially pronounced.

Our key findings are as follows. First, after controlling for various fixed effects, we do not find a consistent and statistically significant relationship between regional temperature and regional per capita income in our panel analysis. Second, in the long-difference analysis, which captures long-term linkages, there is no statistically significant relationship between rising temperatures and changing income levels when analyzing the whole sample of countries. Third, we do not observe a non-linear relationship between temperature and economic development in the short or long run. All these results do not speak to the prevalence of adaptation or intensification effects that matter to all regions in our sample. Fourth, in a threshold approach that considers heterogeneity in the temperature-income relationship, we find no evidence of threshold effects in the short run (fixed-effects panel approach) but do detect them in the long run (long-difference approach). The longrun results support the idea that temperature increases are negatively related to regional economic development only within countries with weak economic-legal and political institutions, leading to intensification effects that reduce long-term regional economic development. Fifth, our findings are robust to alternative operationalizations of regional economic development and warming. Moreover, they also hold when using an alternative economic dataset starting in 1950. Finally, we provide evidence that rising temperatures may be negatively related long-term regional population and education levels within countries with weak economic-legal and political institutions.

Previous research on the determinants of differences in regional economic performance has emphasized the role of within-country differences in, inter alia, geography, human capital (health and education), urban development, entrepreneurial inputs and investment (e.g., Gennaioli et al., 2013, 2014; Cuaresma et al., 2014; Mitton, 2016; Jetter et al., 2019). Given our main empirical findings, we show that regional economic disparities can also emerge from the (long-term) local impacts of global warming especially when accounting for heterogeneity due to country-level differences in economic and institutional vulnerability. This vulnerability emerges as a crucial moderating factor regarding the long-term temperature-income relationship according to our results.

This paper is organized as follows. Section 2 discusses the related literature, while Section 3 describes our regional economic and climate data. Section 4 analyzes the temperature-income relationship using a panel approach. In Section 5 we present findings from a long-difference approach. Section 6 explores short- and long-run heterogeneity in the temperature-income relationship, while Section 7 provides additional robustness checks and empirical extensions. Section 8 explores potential long-run transmission channels. Section 9 concludes.

2. Theory and literature overview

2.1. The temperature-income nexus: Theoretical mechanisms

The literature suggests that higher temperatures could *depress* economic activity through several major pathways. First, higher temperatures may adversely affect *agriculture*, for example, by contributing to water stress or the spread of plant pests (e.g., Deschênes and Greenstone, 2007; Schlenker and Lobell, 2010; Burke and Emerick, 2016; Carter et al., 2018). This may, in turn, adversely affect incomes, especially in economies with large agricultural sectors.

Second, rising temperatures may directly affect *labor productivity* (e. g., Burke et al., 2015a; Letta and Tol, 2019), for example, due to increased heat stress especially if cooling technology such as air conditioning cannot be employed. Such adverse effects on labor productivity may depress industrial and services output, meaning that the adverse economic effects of increasing temperatures would not be restricted to agriculture-dependent economies but also matter to economies that rely more strongly on industrial production and the service industries (e.g., Dell et al., 2014; Carleton and Hsiang, 2016; Nath, 2020).

Third, temperature increases may adversely affect *human capital*. For one, such increases may adversely affect *human health*, for example, by contributing to the spread of disease vectors (e.g., mosquitos that carry malaria or dengue fever) or cardiovascular disease (e.g., Gallup et al., 1999; Barreca, 2012; Deschênes, 2014; Meierrieks, 2021). For another, higher temperatures may also discourage *education*, for example, by contributing to school absenteeism (e.g., Zivin and Shrader, 2016; Zivin et al., 2018; Park, 2022). Consequently, economic activity is expected to suffer as increasing temperatures constrain human capital

accumulation.

Finally, there are further *knock-on effects* that may reinforce the adverse effects of rising temperatures. For instance, by aggravating resource scarcity (e.g., as agricultural land becomes scarcer), temperature increases might promote political instability (e.g., Miguel et al., 2004; Burke et al., 2015b). Political instability, in turn, is expected to depress economic activity. As another potential knock-on effect, by inducing economic and political instability, increasing temperatures may incentivize migration (e.g., Beine and Parsons, 2015; Cattaneo and Peri, 2016; Berlemann and Steinhardt, 2017; Helbling and Meierrieks, 2021). Out-migration may deprive economies of human capital, again depressing economic development.

2.2. Empirical evidence on the temperature-income nexus

Given these theoretical mechanisms, a negative association between higher temperatures and aggregate economic outcomes is the prevailing prior (e.g., Tol, 2009; Carleton and Hsiang, 2016).² Indeed, this prior is mostly consistent with recent empirical studies that suggest that warming may hurt economic performance and is expected to continue to do so as climate change intensifies, especially in already poor and already relatively hot countries. For one, this pertains to empirical studies conducted at the cross-country level (e.g., Tol, 2009; Hsiang, 2010a, 2010b; Dell et al., 2012; Lanzafame, 2014; Burke et al., 2015a, 2018; Letta and Tol, 2019; Kahn et al., 2021). For instance, using a panel of countries from 1960 to 2006, Letta and Tol (2019) show that annual increases in temperature may reduce total factor productivity growth in poor countries, while having no effect in rich countries. For another, studies that examine the relationship between temperature and economic growth within sufficiently large countries tend to come to similar conclusions, e.g., in the cases of China (e.g., Li et al., 2019) or the United States (e.g., Deryugina and Hsiang, 2014; Colacito et al., 2019; Mohaddes et al., 2023). For instance, Mohaddes et al. (2023) analyze the case of 48 U.S. states from 1963 to 2016. They find that climate change has an adverse effect on economic outcomes (e.g., productivity and output) in various U.S. states and economic sectors.

More closely related to our study, a smaller body of empirical research investigates the impact of within-country variation in temperature on sub-national economic outcomes (especially economic growth) with a broader (global) scope. Related studies include Nordhaus (2006), Dell et al. (2009), Zhao et al. (2018), Kalkuhl and Wenz (2020), Greßer et al. (2021) and Kotz et al. (2024). The evidence concerning the temperature-income relationship in these studies is inconclusive. Using cross-sectional data for over 25,000 grid cells (on a $1^\circ~x~1^\circ$ latitude longitude scale), Nordhaus (2006) finds that temperature increases reduce economic activity at the grid level. Zhao et al. (2018) analyze approximately 10,500 grid cells using updated data from Nordhaus (2006) in a panel setting. While they find a negative association between temperature and economic activity, this relationship is statistically significant only in some specifications. Similarly, Dell et al. (2009) study a cross-section of approximately 7500 municipalities in 12 countries in the Americas, showing that while temperature increases were linked to a decline in labor income at the municipal level, this relationship is substantially weaker than any cross-country correlation between temperature and income. Kalkuhl and Wenz (2020) explore sub-national level data from 1900 to 2014 and do not find evidence for temperature effects on permanent growth rates except on the productivity level. Their regional average annual per capita growth rate is 7.0% and, thus, remarkably high in comparison with the average growth rate when

looking at the country level during the same period. Similarly, Greßer et al. (2021) study the relationship between average temperatures and per capita income for a sample of repeated cross-sections of regions, finding no evidence that both variables are related in a statistically significant way. By contrast Kotz et al. (2024) use data for over 1600 regions over 40 years, projecting that the world economy will experience a 19% reduction in income until 2050 independent of future emission choices relative to a world without a temperature increase.

3. Data

3.1. Regional economic development and regional temperature

To empirically investigate the relationship between regional temperature and regional economic development, we draw economic data from the Global Data Lab (Smits, 2016; Smits and Permanyer, 2019; Permanyer and Smits, 2020) as a primary source. This dataset uses data from national statistical offices and various household surveys (e.g., the Demographic and Health Surveys; the UNICEF Multiple Indicator Cluster Surveys; Afrobarometer; or the Integrated Public Use Microdata Series) to provide harmonized sub-national economic data that is comparable across time and space (for a further discussion see Smits, 2016; Smits and Permanyer, 2019; Permanyer and Smits, 2020).³ Our main indicator of regional economic development is the per capita gross national income (GNI) in thousands of 2011 PPP-adjusted US\$. For the Global Data Lab Dataset, "regions" are usually based on official administrative subdivisions used in the countries of interest such as states (e.g., federal states in the United States or Germany), prefectures or districts (Smits, 2016).

Data on our main independent variable, regional temperature, is from a recent update of the *University of Delaware Air Temperature & Precipitation Dataset* of Willmott and Matsuura (2001). This dataset provides data on monthly mean surface air temperatures (available since 1900) at a $0.5^{\circ} \ge 0.5^{\circ}$ grid resolution (approximately 56 km² at the equator).⁴ These temperature values are interpolated for each grid node using data from a set of local weather stations. We use the shape file provided by the *Global Data Lab Dataset* to aggregate the temperature data to the corresponding regional level for which economic data is available. Thus, we have one temperature data-point per year-region observation, allowing us to relate the climate data to the economic data at the regional level.

We can use data for up to 1544 regions in 152 countries. Thus, on average, there are approximately 10 regions per country. A country list is provided in the Appendix. We have available annual data for regional economic development and temperature between 1990 and 2017, where the start and end year of our observation period is dictated by the availability of the economic data.

The summary statistics of our main variables and other explanatory variables employed in our subsequent empirical analyses are reported in Table 1. Focusing on the main dependent variable of interest, there is a large variation in regional per capita income levels. Variation in regional temperatures is substantial, too. Here, the standard deviation associated with regional temperature (7.5 °C) is substantially larger than past global temperature increases (1.09 °C on average) and expected future temperature increases from 2041 to 2060 (1.2 °C to 3.0 °C on average), according to a recent IPCC report (IPCC, 2021).

² A noteworthy recent exception is Zhao et al. (2021) who analyze mortality and ambient temperatures from 750 locations at a grid size of 0.5° x 0.5° across the globe and find that temperatures which minimize mortality are usually well above the median temperature, that is, higher median temperatures can, up to a point, *decrease* mortality.

³ The dataset and information on the methodology can be found at https://globaldatalab.org/ (accessed May 1, 2024).

⁴ A description of this dataset can be found at https://climatedataguide.ucar. edu/climate-data/global-land-precipitation-and-temperature-willmott-mat suura-university-delaware (accessed May 1, 2024).

Summary	v Statistics.
---------	---------------

_						
	Variable	N*T	Mean	SD	Min	Max
	Regional per Capita Income					
	(logged)	40,108	8.88	1.206	5.887	11.55
	Regional HDI	40,108	63.169	17.589	16.8	97.4
	Economic Growth	38,420	0.035	0.082	-1.494	2.737
	Temperature	40,108	18.96	7.558	0.011	31.725
	Maximum Temperature	40,108	27.346	5.197	7.1	42.6
	Temperature Variation	40,108	4.901	3.1	0.198	16.965
	Precipitation (in 10 ml)	40,108	95.54	67.323	0	491.017
	Temperature Deviation					
	from 20-Year Mean	38,296	0.518	1.067	-5.209	10.577

Notes: HDI=Human Development Index. N=Number of regions, T = Number of years. SD=Standard deviation. Summary statistics reported for baseline sample from our primary dataset which excludes regions with temperatures below zero. Temperature variable usually enters models in logged form.

3.2. The temperature-income relationship at the regional level

In Fig. 1, we plot regional per capita income (in logs) against regional temperatures for all regions and years in our primary dataset. The figure illustrates the high variation of the data: Rich regions may experience extreme temperatures, both very hot and cold, like poor regions. This variation suggests that climate alone does not predetermine a region's economic development trajectory, thus challenging notions of climate determinism. However, there is a negative relationship between regional temperatures and regional per capita income, suggesting that warmer regions are poorer when not controlling for other factors that may drive economic development. A fit of a quadratic model (illustrated by the dashed line) performs similarly to the linear counterpart (solid line) in terms of the coefficient of determination. Fig. 1 only reports a simple association between regional temperature and temperature. For instance, we do not account for the time dimension of the data, nor do we account for the role of region- and country-fixed characteristics that may influence the regional temperature-income nexus. We shall do so in subsequent sections.

4. Panel approach

4.1. Empirical strategy employing regional panel data

We analyze the relationship between temperature and per capita income at the regional level by considering the following fixed effects model:

$$Income_{it} = \beta_1 T_{it} + \theta_j + \varphi_{it} + \epsilon_{jit}$$
(1)



Fig. 1. Relationship between Regional Temperature and Per Capita Income.

Here, *Income* refers to the (logged) per capita income of region *j* in country *i* in year *t*. We are most interested in the coefficient β_1 , which reflects the link between (logged) temperature (*T*) and (logged) regional per capita income.⁵ To make statistical inferences, we always compute standard errors that are simultaneously clustered at the region- and parent-country level (Cameroon et al., 2011).

We control for region-fixed effects (θ_i) to account for the role of timeinvariant regional characteristics that may correlate with regional income and temperatures.⁶ Furthermore, we control for time-fixed effects that are interacted with country-fixed effects (φ_{it}). The inclusion of these additional country-time-fixed effects allows us to control for yearspecific effects such as global economic up- and downturns as well as country-specific time trends; country-fixed effects alone do not enter our model as they are perfectly collinear with the region-fixed effects. By accounting for various fixed effects, our baseline specification (1) allows us to investigate changes in GDP per capita between regions, comparing regions that experience higher temperature increases with regions in the same country with lower temperature increases. Country-time-fixed effects allow us to better understand the role of local climate conditions in explaining regional economic disparities because economic effects of climate change that could materialize at the country level are absorbed by the country-year-fixed effects. This, of course, has the important analytical ramification that our empirical results reported below, where our findings are relevant to understanding regional economic responses to weather and climate shocks rather than national socio-economic aggregates such as national economic growth or public health outcomes.

Regarding endogeneity, an effect of regional income on regional temperatures can be plausibly excluded, as temperatures are affected by global anthropogenic and non-anthropogenic factors (e.g., volcanic eruptions). Therefore, it is reasonable to suppose β_1 in Eq. (1) is not affected by reverse causality. Moreover, regional temperature can plausibly be assumed to be external to the regional economy, that is, temperature at the regional level is reasonably regarded as given by economic and political actors. However, this does not imply that temperature is exogenous in an econometric sense (e.g., Deaton, 2010). While our fixed effects strategy captures all regional time-invariant influences (e.g., regional geographic conditions) through θ_i and all country-time variant influences (e.g., national trade patterns over time, national policies over time, etc.) through φ_{it} , there are omitted timevariant variables at the regional level for which we cannot control due to missing data. This may lead to β_1 being biased. As it has become apparent from our previous literature discussion, there is a prevailing prior that temperature increases negatively affect many aspects of human life such as agriculture, health or political stability (e.g., IPCC, 2014). If this prior is correct, by omitting such time-variant regional controls we would overstate any potential negative impact of higher temperatures on regional per capita income. In this sense, we give regional temperatures a comparatively good chance to emerge as a statistically relevant and negative correlate of regional economic development.

⁵ As we use the log of temperature as our explanatory variable, we drop all regions with a negative temperature; this concerns 20 regions (or about 1% of regions in the *Global Data Lab Dataset*) such as Alaska, Russian Siberia as well as parts of Northern Canada and Scandinavia. We do so because our main dependent (economic) variable is also logged. Especially in the long-run, this transformation will help us understand how *growth* in temperature affects *growth* in regional per capita income. As a robustness check, we also use an inverse hyperbolic sine transformation of temperature. Using this alternative transformation does not yield different empirical findings. Finally, below we also use a non-logged temperature variable, showing that our interpretations are not affected by this transformation (see Hsiang, 2010a, 2010b).

⁶ For instance, Jetter et al. (2019) show that access to the sea (by affecting transportation costs) is conducive to regional economic development; at the same time, such geographical features are also expected to influence regional temperature (e.g., as rivers and the sea have cooling effects).

4.2. Empirical results

We report our first set of panel estimates in Table 2. In a setting where no fixed effects are considered (specification 1), we find that higher temperatures are negatively associated regional per capita income at statistically significant levels (p < 0.01). Conditional on separate country- and year-fixed effects (specification 2), however, there is only a marginally significant and negative association between temperature and income at the regional level. Once, we account for regionfixed effects (specification 3), the relationship between temperature and regional income becomes statistically insignificant, with coefficient becoming positive but coefficient sizes approaching zero. The same is true for our baseline specification (specification 4), where we control for regional- and country-year-fixed effects. Interpreting the coefficient for temperature associated with our baseline specification, a 10% increase in regional temperature (approximately 1.9 °C for the sample average), would be associated with a statistically insignificant increase in income per capita of about 0.04%, $CI_{95\%} = [-0.14\%; 0.22\%]$, holding all regional time-invariant and country-year specific conditions constant.

We observe a substantial increase in the goodness of fit (adjusted R^2) when accounting for the various fixed effects. Still, a high R^2 could also indicate that our results are spurious, for example, because both the economic and climate data series are trending in similar ways. To assess this possibility, we always inspect the regression residuals for unit root presence; in case of spurious regression, the residuals would be non-stationary. Using the Fisher-type panel unit root test of Choi (2001), we reassuringly find that the regression residuals are always stationary, dispelling concerns about spurious regression.⁷

As a first robustness check, in specifications 5 and 6 we examine the relationship between temperature and the regional *Human Development Index* (HDI) and the regional growth rate of income, respectively.⁸ We find no statistically robust evidence that temperature is related to any of these alternative economic outcomes.

Finally, we estimate a model where we only focus on regions in Sub-Saharan Africa (specification 7). These regions may be especially vulnerable to rising temperatures, for example, by nature of being in already hot environments or due to the lack of resources to address potential challenges related to rising temperature. Though the coefficient for temperature is negative and larger in absolute terms than for the full sample, it is not statistically significant. Still, the magnitude of the coefficient and its negative sign suggest that regions in countries/ continents that are comparatively poor and potentially vulnerable might potentially experience negative effects of rising temperatures.

To further add to the robustness of our main empirical finding—that temperature increases are on average not associated with changes in regional income—in Table 3 we focus on alternative operationalizations of weather and temperature.

First, Auffhammer et al. (2013: 188) argue that due to the correlation between temperature and *precipitation*, it may be advisable to account for both variables at the same time.⁹ Controlling for (logged) precipitation (drawing data from the *University of Delaware Air Temperature & Precipitation Dataset*), there is still no statistically significant relationship between regional temperature and income. The coefficient size of temperature is close to zero. Precipitation itself is also not a statistically relevant predictor.

Second, we operationalize warming as the maximum temperature per region-year observation to study whether changes in temperature extremes rather than average temperature matter (specification 2); temperature variation indicated by the annual standard deviation of temperature calculated from monthly temperature data to explore the association of regional climate variability and income per capita (specification 3); and temperature in absolute rather than logged form to evaluate whether data transformation matters (specification 5). Regardless of which alternative measure we employ, we continue to find no statistically significant association between any of the measures and regional income. Coefficient sizes are always close to zero.

Third, in the Appendix (Section A1), we also study whether the use of alternative lag structures (e.g., by allowing for deeper lags of temperature) matters to our empirical conclusions. Thereby, we follow empirical contributions that allow temperature increases to take more time to potentially affect GDP per capita or growth (e.g., Kotz et al., 2024).¹⁰ Analyzing different lag structures, we do not find that temperature adversely affects regional income using a variety of lag structures and all coefficients are close to zero.

Finally, there is a discussion in the literature that climatic and economic conditions may be non-linearly related in an inverted U-shaped fashion, where the aggregate economic effects of temperature increases tend to be benign in temperate environments, while temperature increases tend to create adverse effects (e.g., concerning human health, agricultural production, or labor productivity) in already hot environments (e.g., Nordhaus, 2006; Deryugina and Hsiang, 2014; Burke et al., 2015a, 2018; Li et al., 2019). Consequently, we test for a non-linear link between temperature and income via the inclusion of an interacting threshold dummy variable that is equal to unity when mean regional temperatures are larger than 17 °C in specification (5).¹¹ We find no evidence for a non-linear relationship between regional temperature and income. The empirical literature also suggests other temperature thresholds (e.g., Nordhaus, 2006; Deryugina and Hsiang, 2014; Burke et al., 2015a, 2018; Zhao et al., 2018; Li et al., 2019). We therefore consider various alternative thresholds in the Appendix (Section A2). Our results are consistent with the notion that there is no statistically significant non-linear relationship between regional temperature and economic development.

4.3. Discussion

Our panel approach provides no evidence that temperature is systematically and statistically significantly related to regional economic activity after accounting for region-fixed and county-year-fixed effects, meaning that identification of the coefficient for temperature relies on inter-annual fluctuations that are not common across regions of a given

⁷ We use the test of Choi (2001) because it works for the kind of unbalanced panel data we use in this study. Other panel unit root tests require a balanced dataset. We therefore also run our analysis for a fully balanced panel (which covers 1275 regions in 123 countries). We continue to find that temperature is not associated regional income in statistically significant ways. At the same time, we can also inspect the associated regression residuals for non-stationary using more advanced panel unit root tests (which need balanced data) that are robust to heteroskedasticity and cross-sectional dependence. Reassuringly, the panel unit root tests of Herwartz and Siedenburg (2008) as well as Demetrescu and Hanck (2012) also tell us that the associated regression residuals are stationary.

⁸ The sub-national HDI is a translation of the UNDP's official HDI to the regional level, accounting for education (years of schooling), health (life expectancy at birth) and income (see Permanyer and Smits, 2020 for a further discussion).

⁹ Relatedly, Kotz et al. (2022) show that economic growth rates are affected by precipitation, in particular by extreme daily precipitation, but not by daily mean temperatures.

¹⁰ However, in contrast to some of that literature, we do not interact the lags with annual mean temperature over time for our units of observation. Introducing such interactions would also require accounting for interactions with many other regional variables to reduce omitted variable bias potentially affecting the interaction term.

¹¹ Note that the threshold dummy itself is collinear with the fixed effects and is therefore not reported.

Panel Estimates of the Link between Temperature and Regional Income.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Temperature	-1.073***	-0.063*	0.033	0.004	-0.004	-0.005	-0.162
	(0.142)	(0.035)	(0.022)	(0.009)	(0.091)	(0.003)	(0.246)
Country-Fixed Effects	No	Yes	No	No	No	No	No
Year-Fixed Effects	No	Yes	Yes	Yes	Yes	Yes	Yes
Region-Fixed Effects	No	No	Yes	Yes	Yes	Yes	Yes
Country*Year-Fixed Effects	No	No	No	Yes	Yes	Yes	Yes
Panel Unit Root Test (p-value)	(0.00)***	(0.00)***	(0.00)***	(0.00)***	(0.00)***	(0.00)***	(0.00)***
Adjusted R ²	0.260	0.925	0.975	0.997	0.997	0.740	0.991
No. of Observations	40,108	40,108	40,108	40,108	40,108	38,420	10,284
No. of Regions	1544	1544	1544	1544	1544	1536	431
No. of Countries	152	152	152	152	152	151	42

Notes: Dependent variable (DV) is (logged) regional per capita income, except in Models (5) and (6), where it is the regional HDI and regional economic growth rate, respectively. H_0 of the Fisher-type panel unit root test: all panels contain a unit root (i.e., are non-stationary) against the alternative that at least one panel is stationary. Standard errors clustered at the regional and country-year level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table 3

Alternative Operationalizations of Temperature.

	(1)	(2)	(3)	(4)	(5)
Temperature	0.004 (0.009)				
Precipitation	-0.005 (0.004)				
Maximum					
Temperature		-0.002 (0.002)			
Temperature					
Variation			-0.004 (0.003)		
Temperature (not				0.007	
logged)				-0.007	
Temperature (<17 $^{\circ}$ C)				(0.004)	0.007
Temperature (>17 °C)					-0.183 (0.137)
Region-Fixed Effects Country*Year-Fixed	Yes	Yes	Yes	Yes	Yes
Effects	Yes	Yes	Yes	Yes	Yes
Panel Unit Root Test (p-yalue)	(0.00) ***	(0.00) ***	(0.00) ***	(0.00) ***	(0.00) ***
Adjusted R ²	0.997	0.997	0.997	0.997	0.997
No. of Observations	40,107	40,108	40,108	40,108	40,108
No. of Regions	1544	1544	1544	1544	1544
No. of Countries	152	152	152	152	152

Notes: Dependent variable (DV) is (logged) regional per capita income. H₀ of the Fisher-type panel unit root test: all panels contain a unit root (i.e., are non-stationary) against the alternative that at least one panel is stationary. Standard errors clustered at the regional and country-year level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

country. In other words, our findings imply that higher temperatures due to climate change do not necessarily determine economic destiny and development pathways, i.e., there is no evidence for climate determinism. Rather, the results from our panel approach suggest that pathways of regional economic development are largely unrelated to regional temperature, at least in a panel setting that focuses on the shortrun (annual) relationship between both variables.

Clearly, the absence of evidence does not mean evidence of absence of any link between regional temperature and per capita income. That is, climatic conditions may still be linked to regional economic activity. For instance, rather than through deviations from region-specific temperature means or increases in maximum temperatures and temperature variability, adverse economic effects might emerge through extreme weather conditions (e.g., weather-related disasters) which our empirical approach does not fully capture. Nevertheless, it is important to not disregard zero effects of (average) temperature, maximum temperature and temperature variability on regional economic development. Such information on absence of evidence may lead to an update of existing priors (Abadie, 2020), especially given an opinion climate where priors regarding the effects of climate change on indicators of human wellbeing are predominantly negative. Our panel results challenge these existing priors to some extent.

5. Long-difference approach

Global climate change refers to a gradual but non-mean-reverting change in temperatures, meaning that warming becomes more pronounced when longer time horizons are considered (IPCC, 2014, 2021). The cumulative and persistent nature of global climate change may, in turn, be expected to induce cumulative effects on nature and human behavior and, thus, economic outcomes. Such effects may materialize as adaptation or intensification effects (Dell et al., 2014).

Concerning adaptation effects, economic agents may not instantaneously adapt to changing climate conditions. One may also expect adaptive behavior to occur under persistence, that is, after some time has passed; otherwise, adaptive behavior would not be cost-efficient. For example, incentives for farmers to switch to different crops or invest in additional agricultural technology to counter losses in agricultural production are less likely to be economically sound after a short-run but mean-reverting weather shock (e.g., after one particularly hot year) compared with the situation where temperatures do not revert to a stable long-run mean.

Concerning intensification effects, the full adverse effects of rising temperatures may not materialize instantaneously. Rather, effects compound over time. For example, because of persistent warming, in the long run, arable land may permanently vanish due to desertification, salinization or rising sea levels; however, in the shorter run, such effects may remain largely unnoticed.

The presence of adaptation and intensification effects would imply that there are differences between the short- and long-run estimates of regional temperatures on regional per capita income. For instance, if intensification effects matter in the long run, the long-term effects of regional warming on regional economic development may be more pronounced than its short-run effects. This, in turn, might explain the statistically insignificant and quasi-zero relationship between temperature and regional income for a shorter-run time horizon reported in Tables 2 and 3 when accounting for a set of fixed effects.

5.1. Empirical strategy focusing on long-differences

To explore long-run links between rising temperatures and regional per capita income, we resort to the *long-difference approach* applied by

Energy Economics 136 (2024) 107758

Dell et al. (2012, 2014) and Burke and Emerick (2016).¹² This approach involves estimating the following model for region j in country i:

$$\overline{\text{Income}}_{ji,2} - \overline{\text{Income}}_{ji,1} = \alpha + \beta_1 \left[\overline{T}_{ji,2} - \overline{T}_{ji,1} \right] + \varphi_i + \epsilon_{ji}.$$
(2)

Here, we first construct region-specific (logged) averages in per capita income and temperature between 1990 and 1993 (subscript 1 in ji, 1) and between 2014 and 2017 (subscript in ji, 2). Then, we take the so-called long-difference associated with these variables: that is, we subtract these averages from each other. This allows us to gauge the extent of regional economic development and warming between the early 1990s and mid-2010s. The two periods are chosen for two reasons. First, in this manner we can maximize the temporal differences between both data points, making it more likely that we can indeed capture the adverse economic effects of climate change that are expected to emerge especially over longer time horizons. Second, by taking averages over multiple years we can reduce the influence of short-run temperature and business cycle fluctuations.

Fig. 2 illustrates that most regions indeed experienced some warming between the 1990–1993 and 2014–2017 periods, where the average regional level of warming in our sample was 0.78 °C, which is broadly consistent with recent IPCC reports for global land temperature increases (IPCC, 2014, 2021). Variation in temperature changes is, at the same time, substantial.

In general, estimating Eq. (2) allows us to evaluate how differences in temperature between 1990–1993 and 2014–2017 (which are indicative of non-mean-reverting warming) are related to differences in regional economic development over the same time periods. Importantly, our use of regional data still allows us to include a set of country-fixed effects (φ_i), thereby again improving on the cross-country literature. For instance, country-fixed effects account for initial country-wide temperature levels, that is, they account for the fact that regions are either located in a generally warm or cold country. The constant (α) accounts for trending in the dependent variable between the *early* (subscript 1) and the *late* (subscript 2) period. As the long-difference approach requires data for both the late and early period, we run this analysis for a subsample of approximately 1300 regions in 125 countries. We compute heteroskedasticity-robust standard errors to make statistical inferences.

The long-difference approach complements our previous panel analysis, as also previously summarized by Dell et al. (2014) and Burke and Emerick (2016). For one, given that we estimate the potential economic effects of regional warming from long-run changes in average climate conditions rather than short-run annual variation (as we did in the panel approach), the long-difference approach is less susceptible to extreme (but mean-reverting) temperature events and more likely to capture a potential impact of (non-mean-reverting) climate change. Therefore, the long-difference approach is closer to identifying long-run relationships between temperature and income accounting for adaptation or intensification effects that only materialize over longer time horizons (e.g., Dell et al., 2014: 778). At the same time, we can directly compare how regional economic development is linked to short-run (panel approach) and long-run temperature variation (long-difference approach).¹³ Thus, we can quantify whether long-run adjustment to rising temperatures (in terms of regional per capita GDP) is smaller (consistent with adaptation) or larger (consistent with intensification) than short-run adjustment (Dell et al., 2012, 2014; Burke and Emerick, 2016).

5.2. Empirical results

Our long-difference estimates of Eq. (2) are reported in Table 4. Briefly summarized, the findings do not indicate that temperature increases between the 1990–1993 versus 2014–2017 periods are robustly associated with lower levels of regional economic development at conventional levels of statistical significance. Coefficients are positive or negative depending on the specification but their sizes for the change in regional temperate tend to be close to zero. Thus, the long-difference estimates speak to our panel results of Tables 2 and 3 in that we find little robust evidence of an unfavorable effect of temperature increases on regional economic outcomes.

In detail, in both a parsimonious model (specification 1) and a model controlling for rainfall (specification 2), a change in regional temperature is positively but statistically insignificantly related with changes in regional incomes. When we use alternative temperature measures (specifications 3–5), the positive temperature coefficient turns negative and statistically significant at the 10% level for the association between long-run increases in maximum temperatures and changes in regional income, while the association between income and non-logged temperature and temperature variation, respectively, is not statistically significant. There is also no evidence for a non-linear relationship (specification 6). While long-differences in temperature share no robust relationship with changes in the regional HDI (specification 7), they are negatively associated with changes in regional economic growth (specification 8) at the 10% significance level. Finally, when focusing only on regions in Sub-Saharan Africa (specification 9), we do not find evidence for a statistically robust negative relationship when employing the long-difference approach, though, the coefficient size increases in absolute terms.

6. Heterogeneity in the regional temperature-income relationship

6.1. Empirical strategy to explore heterogeneity

So far, both our panel and long-difference analyses suggest that regional economic development is not (adversely) affected by regional temperature shocks. In other words, comparing regions that experience higher temperature increases with regions in the same country with lower temperature increases (and controlling for a variety of fixed effects), the former regions do not exhibit stronger adverse economic effects due to warming. This, in turn, suggest that there is no uniform effect of warming on regional economic disparities within countries. One theoretical argument to explain this finding is that any regional disparities due to warming are offset at the national level, e.g., by government activity or adaptive behavior of economic agents. At the same time, however, this argument would also suggest that there could be heterogeneity in the temperature-income relationship, with regions being economically exposed to temperature increases when they are in countries that lack the capability to offset regional disparities. Thus, below we explore various country-specific conditions that may make it more or less likely for a link between temperature and income to emerge.

Indeed, the cross-country literature suggests that certain *country-specific conditions* may moderate the temperature-income relationship. Most prominently, existing research suggests that a country's income level matters. Here, it is argued that poor countries may lack the adaptative capability to counter the adverse effects of weather shocks or warming and are thus expected to suffer more adverse economic effects (e.g., Dell et al., 2012, 2014; Burke et al., 2015a).

Similarly, such conditions might also affect a region's vulnerability to rising temperatures. Below, we consider the role of country-specific economic, legal and political conditions in the short and long run to investigate potential heterogeneities in the temperature–income relationship. Here, we first consider a panel threshold-model of the

¹² For other approaches to differentiating short and long-run links between temperature and income, see for example, Deryugina and Hsiang (2017), Lemoine (2021) or Kahn et al. (2021).

¹³ This is because the long-difference approach in Eq. (2) is equivalent to the panel approach of Eq. (1), with the difference between both approaches being that Eq. (2) is not expressed in years but in decades (see also Dell et al., 2014: 778).



Fig. 2. Histogram of Change in Temperature (1990–93 versus 2014–2017).

Table 4 Long-Difference Estimates for the Link between Temperature and Regional Incomes.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Δ Temperature	0.014 (0.109)	0.014 (0.110)					-0.002 (0.010)	-0.048* (0.026)	-1.259 (0.833)
Δ Precipitation		-0.007 (0.039)							
Δ Maximum Temperature			-0.016* (0.009)						
Δ Temperature Variation				-0.046 (0.030)					
Δ Temperature (No Log)					-0.026* (0.013)				
Δ Temperature (<17 $^\circ\text{C})$						0.019 (0.108)			
Δ Temperature (>17 $^\circ\text{C})$						-0.013 (0.108)			
Country Dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Period Dummy (Intercept)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Adjusted R ²	0.996	0.996	0.996	0.996	0.996	0.996	0.996	0.927	0.989
No. of Observations	1288	1288	1288	1288	1304	1288	1288	1288	227
No. of Regions	1288	1288	1288	1288	1304	1288	1288	1288	277
No. of Countries	125	125	125	125	125	125	125	125	31

Notes: Dependent variable (DV) is difference of the (logged) regional per capita income the early and late period (1990–1993 vs. 2014–2017), except in Models (7) and (8), where it is the regional HDI and regional economic growth rate, respectively. Δ always refers to the difference between the early and late period (1990–1993 vs. 2014–2017). Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

following form:

$$Income_{jit} = \beta_1 T_{jit-1}(q_{it} < \gamma) + \beta_2 T_{jit-1}(q_{it} \ge \gamma) + \theta_j + \varphi_{it} + \epsilon_{jit}$$
(3)

In the long-difference setting, this model has the following form:

$$\overline{\text{Income}}_{ji2} - \overline{\text{Income}}_{ji1} = \beta_1 \left[\overline{T}_{ji2} - \overline{T}_{ji1} \right] (q_{it} < \gamma) + \beta_2 \left[\overline{T}_{ji2} - \overline{T}_{ji1} \right] (q_{it} \ge \gamma)$$
$$+ \alpha + \varphi_i + \epsilon_{ji}$$
(4)

In both the short- and long-run case, the threshold model follows Hansen (1999). Here, the threshold parameter γ divides the respective equation into two regimes that describe the effect of temperature on income below and above the threshold. The threshold describes a structural break in the relationship between temperature and income. For instance, in poor countries (below a certain country-level income threshold) a potential link between rising temperatures and regional economic development might be more pronounced than in comparatively rich countries (above the income threshold). The exact value of γ is determined empirically following Hansen (1999). Testing for a threshold effect is the same as testing for the equality of coefficients

between both regimes (i.e., to test whether $\beta_1 = \beta_2$ for both equations). Rejecting the null hypothesis of equal coefficients would imply that the threshold approach is more informative than the non-threshold models estimated before. Moreover, in such a case specific country-wide conditions can explain the emergence of a link between temperature and income.

6.2. Threshold variables

We explore potential country-specific threshold variables that may account for differential associations between rising regional temperatures and regional per capita income in both the short and long run. These country-specific economic-legal and political variables may, in turn, be affected by rising temperatures and thus potentially be endogenous. For instance, Brückner and Ciccone (2011) find that changing weather conditions may foster democratic governance by lowering the opportunity cost of contesting autocratic power. To address such endogeneity concerns, we focus on *initial* economic-legal and politicoinstitutional conditions (as averages over the 1980–1989 period). In detail, we consider the following six variables:

- 1. **Per capita** *GDP*: Per capita *GDP* at the country-level is drawn from the *World Development Indicators* (World Bank, 2021). Employing this variable as a moderator speaks to the idea that poorer countries may be more vulnerable to adverse consequences of global climate change (e.g., Dell et al., 2012; Burke et al., 2015a). For instance, they may lack the resources to invest in technology (e.g., agricultural machinery) and public goods (e.g., levees) to counter unfavorable warming effects on their economies.
- 2. *Democracy*: Democratic development is indicated by an index of *electoral democracy* from the *V-DEM Dataset* of Coppedge et al. (2021). Regions within non-democratic countries may be more vulnerable to the impact of rising temperatures. For instance, non-democratic governments may be less likely to respond to climate change by adjusting public policy because they do not depend on electoral consent for political survival.
- 3. Civil Liberties: An index of equality before the law and individual liberty from the V-Dem Dataset accounts for a broad range of political and legal-economic civil liberties (e.g., property rights protection, access to the justice system and freedom of movement). By accounting for legal and economic liberties, this variable is distinct from the democracy variable, reflecting the broader institutional framework that would allow economic agents to adequately respond to warming to mitigate its economic effects. For instance, sound legal and political institutions may encourage private (long-run) investment and innovation because they promote private contracting and provide checks against expropriation (e.g., North, 1981; Acemoglu and Johnson, 2005). Investment and innovation, in turn, are potentially relevant to reducing vulnerability to rising temperatures. For instance, private businesses are more likely to invest in measures that reduce their vulnerability to rising temperatures (e.g., air conditioning, flood walls and supply line security) when the risk of expropriation and predation is low.
- 4. *Equality*: An index of *egalitarianism* from that the *V-Dem Dataset* considers the extent of equality of access to rights, freedoms, public goods and political power between different societal groups. Potentially, higher levels of equality reduce vulnerability to the adverse economic effects of rising temperatures by providing vulnerable segments of society (e.g., the poor) with resources (e.g., access to public health) to counter them.
- 5. *Rural Exclusion*: Potential adverse economic consequences of warming could be more strongly felt in countries that disfavor their rural parts. For example, when climate change threatens agricultural production, but a national politics prioritizes urban over rural areas, this may exacerbate related economic losses. We use an index of *rural exclusion* (accounting for, e.g., differences in political power and access to public goods between cities and rural areas) from the *V-Dem Dataset* to account for this idea.
- 6. *Composite Measure*: Comparatively rich countries tend to be more democratic, while democratic countries, in turn, tend to promote equality and civil liberties. Thus, we also construct a *composite measure of sound economic and institutional starting conditions* by means of *principal component analysis*. Principal component analysis is used to reduce the dimensionality of a dataset with many interrelated variables, while retaining as much information and variation

as possible (e.g., Jolliffe, 2002). We extract the first principal component as our composite measure. $^{14}\,$

In line with our discussion above and prevailing priors, we expect countries with relatively sound economic and institutional starting conditions to be less vulnerable to the adverse economic consequences of rising temperatures.

6.3. Empirical results: Heterogeneity in the short and long run

We report our panel threshold estimates in Table 5 and our longdifference threshold estimates in Table 6. For each threshold variable, we identify a likely threshold value following Hansen (1999). For instance, for the panel approach (Table 5) countries with a 1980–1989 per capita income below 994 US\$ would be considered as relatively poor; this concerns 33 countries, mainly located in Sub-Saharan Africa and parts of Asia.

Estimating the panel threshold model, we do not find any heterogeneous link between temperature and regional per capita income (specification 1). For regions in poor countries the coefficient between temperature and income is positive, while it is negative for regions in rich countries. Both coefficients are close to zero and statistically not different from zero at conventional significance levels. All other panel threshold models (specifications 2-6) yield similar interpretations: The estimated coefficients are not statistically significant and rather close to zero, suggesting no heterogeneity in the link between temperature and regional per capita income in the short run. Testing for the equality of coefficients below and above the various estimated thresholds, we also find little evidence threshold effects to matter. Thus, we find no evidence that regional temperatures affect regional economic development in the panel threshold approach, regardless of which threshold variable we consider. This provides indirect support for our more parsimonious panel models reported above suggesting no general link and no heterogenous link between temperature and income in the short run.

In Table 6, we consider heterogeneity in the temperature-income relationship in the long run. We do not find any statistically significant heterogeneity in the long-run link between regional temperature and income for regions in poor vs. rich countries (specification 1) and democratic vs. not democratic countries (specification 2). Still, some heterogenous relationships emerge. We find evidence that temperature is associated with long-run regional economic development in countries with relatively poor (initial) economic-legal and political conditions, characterized by relatively weak civil liberties (specification 3), low levels of equality (specification 4), strong rural exclusion (specification 5) as well as less sound starting conditions as measured by our composite measure (specification 6). By contrast, for regions in countries that have strong (initial) civil liberties, high levels of equality, weak rural exclusion and sound starting conditions no statistically significant relationship between temperature and per capita income emerges in the long run.

Focusing on our composite measure, our long-difference threshold estimates imply that the adverse effect of temperature increases concern 350 regions in 17 countries that have unsound starting conditions such as Haiti, Laos, Nepal and the Democratic Republic of the Congo. For regions in these countries, we find that a 10% increase in temperature is associated with a substantial decrease in per capita income by

 $^{^{14}}$ The factor loadings for the first principal component are 0.43 (per capita income), 0.46 (democracy), 0.45 (civil liberties), 0.43 (equality) and -0.46 (rural exclusion), implying that higher values of the composite measure correspond to higher income levels, stronger democratic development and civil liberties, more equal institutions and lower levels of rural exclusion. The Kaiser-Meyer-Olkin measure of sampling adequacy associated with the principal component analysis is 0.82, indicating that the results of the analysis are meritorious (e.g., Jolliffe, 2002).

Panel Threshold Estimates.

	(1)	(2)	(3)	(4)	(5)	(6)
Moderator \rightarrow	Per Capita Income	Electoral Democracy	Civil Liberties	Equality	Rural Exclusion	Composite Measure
Interpretation when Moderator = 1	Rich	Democratic	Relatively Free	Relatively Equal	Strong Rural Exclusion	Sound Starting Conditions
Temperature (Moderator $= 0$)	0.033	0.006	0.014	-0.122	0.004	0.009
	(0.022)	(0.035)	(0.029)	(0.110)	(0.023)	(0.032)
Temperature (Moderator $= 1$)	-0.040	-0.023	-0.044	0.018	-0.049	-0.022
	(0.035)	(0.029)	(0.033)	(0.020)	(0.060)	(0.031)
[Equality of Coefficients Test p-value]	[0.08]*	[0.51]	[0.19]	[0.21]	[0.41]	[0.49]
Threshold Estimate	994\$	0.16	0.47	0.41	0.54	-1.39
Region-Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
Country*Year-Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
Adjusted R ²	0.996	0.996	0.996	0.997	0.996	0.996
No. of Observations	31,052	33,740	33,740	33,740	33,180	30,156
No. of Regions	1109	1205	1205	1205	1185	1077
No. of Countries	102	112	112	112	111	98

Notes: Dependent variable is (logged) per capita income. Threshold estimates for Models (2) to (5) refer to values of respective index. Standard errors clustered at the regional and country-year level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table 6

Long-Difference Threshold Estimates.

	(1)	(2)	(3)	(4)	(5)	(6)
Moderator (=Threshold Variable) \rightarrow	Per Capita Income	Electoral Democracy	Civil Liberties	Equality	Rural Exclusion	Composite Measure
Interpretation when Moderator = 1	Rich	Democratic	Relatively Free	Relatively Equal	Strong Rural Exclusion	Sound Starting Conditions
Δ Temperature (Moderator = 0)	-0.462	-0.161	-0.190*	-1.144***	0.097	-1.088**
	(0.371)	(0.101)	(0.106)	(0.412)	(0.092)	(0.488)
Δ Temperature (Moderator = 1)	-0.070	0.153	0.154	0.122	-1.196**	-0.060
	(0.104)	(0.098)	(0.095)	(0.086)	(0.581)	(0.114)
[Equality of Coefficients Test p-value]	[0.31]	[0.02]**	[0.01]**	[0.00]***	[0.02]**	[0.04]**
Threshold Estimate	1663\$	0.25	0.32	0.40	0.66	-1.57
Country Dummies	Yes	Yes	Yes	Yes	Yes	Yes
Period Dummy (Intercept)	Yes	Yes	Yes	Yes	Yes	Yes
Adjusted R ²	0.995	0.996	0.996	0.996	0.996	0.995
No. of Observations	1118	1216	1216	1216	1196	1084
No. of Regions	1118	1216	1216	1216	1196	1084
No. of Countries	104	113	113	113	112	99

Notes: Dependent variable (DV) is difference of the (logged) per capita income between the early and late period (1990–1993 vs. 2014–2017). Δ refers to the difference between the early and late period (1990–1993 vs. 2014–2017). Threshold estimates for Models (2) to (5) refer to values of respective index. Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

approximately 10.9%, $CI_{95\%} = [-1.3\%; -20.5\%]$. For the remainder of the sample of regions in all other countries, temperature is not found to sway regional economic development in the long run in statistically robust ways.

As argued by Dell et al. (2012, 2014), the long-difference estimates can be interpreted as capturing adaptation or intensification effects. According to this interpretation, for regions within especially vulnerable countries, intensification effects (where the long-difference estimates are more pronounced than their panel counterparts) appear to matter. For instance, this may point to the role of rising temperatures in hurting regional economic development in the long run when weak institutions do not sufficiently incentivize investment, innovation and other forms of adaptation. At the same time, it is important to note that long-run changes in temperature may be linked to a variety of unobservables that could be relevant to regional economic development, too. That is, if higher temperatures negatively affect other unobserved factors which, in turn, affect regional income, our estimates may constitute an upper bound of the long-run economic impact of warming in regions within especially vulnerable countries. For instance, if political instability is such an unobserved factor, improving political stability (e.g., by fostering conflict prevention) may contribute to lowering the adverse effects of climate change in regions within particularly vulnerable countries.

7. Further robustness checks and empirical extensions

7.1. Use of alternative economic data

The use of alternative data sources may matter to empirical analyses (see, e.g., Johnson et al., 2013 with respect to the use of GDP data). Thus, as part of our robustness checks, we employ alternative data for regional GDP per capita from Gennaioli et al. (2014) to cross-validate our main empirical findings (see also Greßer et al., 2021). Drawing from national and regional statistical offices, Gennaioli et al. (2014) report regional economic activity as *regional* per capita *GDP*.¹⁵ Here, "region" refers to the "most disaggregated administrative division available (typically states or provinces), or, when such data does not exist [...] the most disaggregated statistical division level" (Gennaioli et al., 2014: 266).

The dataset of Gennaioli et al. (2014) allows us to consider 1446 regions in 81 countries between 1950 and 2014. Thus, we can extend the temporal coverage of the data, while the number of countries is reduced. The country list in the appendix details the country coverage. The economic data are not observed annually and Gennaioli et al. (2014) do not

¹⁵ To make the data comparable between regions and countries, the data is provided as per capita GDP in constant 2005 purchasing power parity dollars. For further information on their methods, we refer to Gennaioli et al. (2014).

provide external data (e.g., on different growth determinants) that could be used to reliably interpolate the missing data. Following Gennaioli et al. (2014: 266) and, more generally the past growth literature, we thus analyze a series of five-year period averages (1950–54, 1955–59, etc.). This allows us to consider a maximum of 13 consecutive five-year observations per region, potentially making it more likely to uncover—given the longer time horizon—unfavorable long-run effects of rising temperatures. On average, we observe approximately six five-year periods per region, meaning that the panel dataset is unbalanced (see also Gennaioli et al., 2014: 268–270).

We estimate the relationship between the alternative regional economic development variable and temperature in both a panel and longdifference setting. To maximize coverage, in the long-difference setting the early period is the 1960–1980 period, which is compared (by means of taking the long-difference) to the late period of 1990–2010. In both settings, we also allow for heterogeneous effects by amending our models with a variable that is equal to unity when economic and institutional conditions are sound and zero otherwise, using the composite measure (from a PCA analysis) of the soundness of country-specific economic and institutional circumstances (with respect to per capita income, democracy, equality etc.) that we already constructed above as the moderator variable.

We report our results in Table 7. We find that higher temperatures are not associated with lower per capita GDP levels both in the homogenous and heterogeneous panel setting. In the long-difference

Table 7

Use of Alternative Economic Data.

	(1)	(2)	(3)	(4)
Empirical Approach \rightarrow	Panel Approach		Long-Diffe Approach	erence
Dependent Variable \rightarrow	Regional _J GDP	per Capita	Regional j GDP	per Capita
Temperature	-0.028 (0.030)			
Temperature (Moderator = 0)		-0.038 (0.055)		
Temperature (Moderator = 1)		-0.013 (0.034)		
Δ Temperature			-0.097 (0.062)	
Δ Temperature (Moderator = 0)				-0.185*** (0.049)
Δ Temperature (Moderator = 1)				0.125 (0.138)
Region-Fixed Effects	Yes	Yes		
Country*Period-Fixed Effects	Yes	Yes		
Country Dummies			Yes	Yes
Period Dummy (Intercept)			Yes	Yes
Adjusted R ²	0.982	0.984	0.981	0.988
No. of Observations	8793	7537	675	560
No. of Regions	1446	1133	675	560
No. of Countries	81	61	81	67

Notes: Dependent variable (DV) is (logged) regional per capita GDP. Δ always refers to the difference between the early and late period (1960–1980 vs. 1990–2010). "Moderator" refers to composite measure measuring economic and institutional conditions for the 1980–1989 period as described in the main text, where "Moderator = 1" refers to sound conditions. Standard errors clustered at the regional and country-year level in parentheses for Models (1) to (3). Robust standard errors in parentheses for Models (4) to (6). *** p < 0.01, ** p < 0.05, * p < 0.1.

approach, temperature increases are not associated to changes in regional per capita GDP. However, we observe a negative and statistically significant association between temperature and per capita GDP in the long run for regions in countries with comparatively weak economic and institutional country-specific conditions.¹⁶ Thus, when using the data of Gennaioli et al. (2014) we arrive at empirical findings that mirror those reported above where we employed the economic data from the *Global Data Lab Dataset* such that our overall interpretations do not change: There is no evidence of a homogeneous or heterogeneous effect of rising temperatures on regional economic development in the short run when accounting for different fixed effects. For regions in countries with a weak economic, legal and political institutions, the association between higher temperatures and regional income is negative in the long run.

7.2. Warming as deviations from long-run temperature means

As another robustness check, we consider an alternative way to investigate potential effects of increasing temperatures, by analyzing the annual deviation of temperature from average temperatures observed in the 20 years prior. We calculate these temperature deviations as described in Kahn et al. (2021: 4). Kahn et al. (2021) argue that operationalizing warming in this manner may avoid introducing a linear trend in temperature (as it follows from global warming) and thus also avoid a spurious regression problem. A similar approach to measuring warming is also used by Mohaddes et al. (2023).

We consider the relationship between regional economic development and warming—measured as deviations from long-term temperature means—in both a panel and long-difference setting with our primary economic dataset. Our findings are reported in Table 8. We find no evidence that year-to-year temperature deviations matter to regional per capita income (panel approach). In the long run, temperature deviations reduce economic activity only in regions within countries with poor starting conditions, i.e., within countries that are especially vulnerable. Thus, the findings in Table 8 are fully consistent with our main empirical findings—where we consider (logged) temperature as our baseline climate measure—reported above.

7.3. Capturing long-run linkages through a moving-difference approach

To further explore potential long-run linkages between temperature and income, we perform a robustness test by employing a movingdifference approach.¹⁷ We start by calculating the five-year difference between the economic and temperature variables, respectively. For instance, this means we calculate the difference in regional per capita income between 1990 and 1995, 1991 and 1996, 1992 and 1997 and so on. We then estimate how five-year differences in temperature are associated with five-year differences in per capita income at the regional level. The advantage of this approach over the long-difference approach employed above is that we can still exploit the time-series dimension of the data to some extent, allowing us to employ a larger number of observations and increasing variation in the data. At the same time, it comes at the potential expense of short-run temperature and business cycle fluctuations becoming more influential, which are averaged out in the long-difference approach employed above. When applying the moving-difference approach, we also consider longer time horizons by means of constructing ten-year and fifteen-year moving averages, respectively, to account for the (theoretical) argument that the economic effects of higher temperature might only be felt after some time

¹⁶ Given that the country-specific data is not available for all countries, when studying heterogeneous effects, our sample shrinks. We therefore also estimate the homogenous panel and long-difference estimates for a reduced sample. We do not find that our results are driven by sample choices.

¹⁷ We thank an anonymous reviewer for pointing out this empirical approach.

Warming as Deviations from Long-Run Means.

	(1)	(2)	(3)	(4)
Empirical Approach \rightarrow	Panel Ap	proach	Long-Diff Approach	erence
Dependent Variable \rightarrow	Regional Capita GI	per DP	∆ Region Capita GI	al per DP
Temperature Deviation	-0.001 (0.002)			
Temperature Deviation (Moderator				
= 0)		0.001		
		(0.003)		
Temperature Deviation (Moderator				
= 1)		-0.003		
		(0.003)		
Δ Temperature Deviation			-0.007	
			(0.007)	
Δ Temperature Deviation				
(Moderator $= 0$)				-0.046**
				(0.019)
Δ Temperature Deviation				
(Moderator $= 1$)				-0.003
				(0.008)
Region-Fixed Effects	Yes	Yes		
Country*Period-Fixed Effects	Yes	Yes		
Country Dummies			Yes	Yes
Period Dummy (Intercept)			Yes	Yes
Adjusted R ²	0.996	0.996	0.995	0.995
No. of Observations	38,761	29,176	1252	1042
No. of Regions	1480	1042	1252	1042
No. of Countries	151	99	125	99

Notes: Dependent variable (DV) is (logged) regional per capita GDP. Δ refers to the difference between the early and late period (1990–1993 vs. 2014–2017). "Moderator" refers to composite measure measuring economic and institutional conditions for the 1980–1989 period as described in the main text, where "Moderator = 1" refers to sound conditions. Standard errors clustered at the regional and country-year level in parentheses for Models (1) to (3). Robust standard errors in parentheses for Models (4) to (6). *** p < 0.01, ** p < 0.05, * p < 0.1.

has passed.

We report our moving-differences findings in Table 9. We find that rising temperatures are not statistically significantly associated with changes in regional income when we consider five-year moving averages (positive coefficient estimates of temperature but close to zero) or ten-year moving differences (negative coefficient estimates of temperature but close to zero) in both a homogeneous and heterogeneous setting. However, a statistically significant and negative relationship emerges when we consider fifteen-year moving averages, where we find-on closer inspection-that this relationship is especially felt in regions within countries with poor starting conditions.¹⁸ That is, we find the usual evidence for heterogeneity in the economic response to rising temperatures. Finding that this relationship is most precisely estimated when considering fifteen-year moving averages suggests that any potential adverse economic effects of warming are more likely to emerge (and thus be empirically detected) in the longer run, where they are limited to regions in particularly vulnerable countries.

7.4. Further sources of country-level heterogeneity

When studying heterogeneity in the temperature-income relationship, above we focused especially on the role of country-specific institutional conditions. Clearly, however, there are many other conditions or variables that may also moderate the relationship between regional temperature and income. Thus, as an empirical extension, in this sub-section we explore four additional sources of heterogeneity: a country's dependence on agricultural and industrial production, respectively, its country-level exposure to temperature changes as well as its response to the economic shocks of warming at the country level. In the Appendix (Section A3) we discuss on more detail the theoretical rationale behind each moderator and how we operationalize each moderator.

We report our empirical results in Table A3 in the Appendix. In short, we find that local temperature increases do not matter to regional income in the short run but only in the long run in those regions that are especially vulnerable to rising temperatures as operationalized by the country-specific dependence on manufacturing and the country-specific (i.e., national average) experience of warming. These findings are, of course, in line with our main results reported above. In detail, we find weak evidence that higher temperatures reduce regional economic development in the long run especially when a country's manufacturing sector is small. Furthermore, we show that regional temperature increases are negatively associated with regional economic activity in the long run when the region of interest is in a country that—as a whole--has not experienced high levels of warming. This finding may imply that such a region receives fewer resources from the central government to adapt to higher temperatures because the central government sees less necessity or has fewer incentives to foster adaption. This interpretation of the heterogenous link in the long run would again speak to our idea and the evidence presented before that institutional arrangements at the country level matter to the regional temperature-income nexus. At the same time, however, future research is necessary to examine the country-specific roots of heterogeneity in this nexus in more detail.

8. Exploring long-run transmission channels

Finally, we explore potential mechanisms through which higher temperatures may hurt regional long-run economic development in vulnerable countries. We focus on three potential mechanisms: (1) regional population size, (2) regional levels of education (in years of schooling) and (3) regional health (measured as life expectancy at birth). All three variables come from the *Global Data Lab Dataset*.

For one, higher temperatures may adversely affect regional population growth, e.g., by influencing fertility decisions or inducing outmigration (e.g., Lam and Miron, 1996; Beine and Parsons, 2015; Berlemann and Steinhardt, 2017; Barreca et al., 2018; Sellers and Gray, 2019; Helbling and Meierrieks, 2021). For another, higher temperatures may depress human capital (education and health) by contributing to the spread of diseases or malnutrition (e.g., Barreca, 2012; Deschênes, 2014; Zivin and Shrader, 2016; Zivin et al., 2018; Meierrieks, 2021; Park, 2022). The reduction in the availability of (skilled and healthy) human labor due to rising temperatures may, in turn, reduce regional economic development (e.g., Weil, 2013; Gennaioli et al., 2013, 2014).

Table 10 shows that higher temperatures do not have a uniform and statistically significant association with the three potential transmission channels in the long run (specifications 1, 3 and 5). However, in the long run higher temperatures are negatively associated with population size and education in those regions within countries characterized by poor initial economic-legal and political conditions (specifications 2 and 4); there are no comparable associations for human health measured by life expectancy (specification 6). That is, higher temperatures are only adversely associated with potential determinants of economic development precisely in those regions for which we also find higher temperatures being negatively associated regional economic development in the long run.

There are two caveats to this exploration of transmission channels. First, we only have available data for some specific indicators of population and human capital development as provided by the *Global Data*

¹⁸ As sample size is reduced when we consider longer moving differences, we also test whether our findings are due to changes in sample size by estimating specifications (1) and (3) with the same observation as specification (5) and estimating specifications (2) and (4) with the same observations as specification (6). Results for five- and ten-year moving averages remain statistically insignificant.

Capturing Long-Run Linkages: Moving-Difference Approach.

	(1)	(2)	(3)	(4)	(5)	(6)
Moving-Differences Time Horizon \rightarrow	5 Years		10 Years		15 Years	
Temperature	0.009		-0.003		-0.009*	
	(0.006)		(0.008)		(0.005)	
Temperature (Moderator $= 0$)		0.006		-0.017		-0.024**
		(0.032)		(0.027)		(0.011)
Temperature (Moderator $= 1$)		0.009		0.023		0.008
		(0.045)		(0.046)		(0.018)
Region-Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
Country*Year-Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
Adjusted R ²	0.895	0.859	0.935	0.918	0.975	0.970
No. of Observations	32,370	24,771	24,740	19,386	17,254	14,001
No. of Regions	1539	1077	1523	1077	1428	1077
No. of Countries	152	98	150	98	139	98

Notes: Dependent variable (DV) is (logged) regional per capita income. "Moderator" refers to composite measure measuring economic and institutional conditions for the 1980–1989 period as described in the main text, where "Moderator = 1" refers to sound conditions. Standard errors clustered at the regional and country-year level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table 10

Exploration of Long-Run Transmission Channels.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent Variable →	Δ Population		Δ Mean Years of S	chooling	Δ Life Expectancy a	at Birth
Δ Temperature	0.006		-0.203		0.364	
	(0.129)		(0.195)		(1.155)	
Δ Temperature (Moderator = 0)		-1.232^{**}		-4.506***		1.351
		(0.610)		(1.405)		(7.110)
Δ Temperature (Moderator = 1)		0.108		-0.067		1.682
		(0.160)		(0.349)		(1.941)
[Equality of Coefficients Test p-value]		[0.03]**		[0.00]***		[0.96]
Country Dummies	Yes	Yes	Yes	Yes	Yes	Yes
Period Dummy (Intercept)	Yes	Yes	Yes	Yes	Yes	Yes
Adjusted R ²	0.993	0.993	0.994	0.994	0.989	0.988
No. of Observations	1282	1084	1288	1084	1288	1084
No. of Regions	1282	1084	1288	1084	1288	1084
No. of Countries	124	99	125	99	125	99

Notes: Δ always refers to the difference between the early and late period (1990–1993 vs. 2014–2017). "Moderator" refers to composite measure measuring economic and institutional conditions for the 1980–1989 period as described in the main text, where "Moderator = 1" refers to sound conditions. Robust standard errors in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Lab. Other variables measuring, e.g., urbanization, tertiary education or child mortality may exhibit a different relationship with rising temperatures, while also sharing additional links with regional economic development. Second, due to data constraints, our exploration of potential transmission channels cannot account for further potential transmission channels from rising temperatures to reduced regional economic activity. For instance, this includes regional measures of political instability, resource scarcity, labor productivity or agricultural production. Indeed, there is ample evidence that these variables also matter to the temperature–income nexus as relevant transmission variables (e.g., Schlenker and Lobell, 2010; Dell et al., 2012, 2014; Burke et al., 2015b; Carter et al., 2018).

9. Conclusion

Motivated by growing concerns about the adverse effects of rising temperatures on human well-being and economic prosperity, we study the relationship between temperature and per capita income using annual sub-national data for over 1500 regions in 152 countries between 1990 and 2017. In so doing, we add to the currently still small body of literature on the economic consequences of climate change at the sub-national level as well as to the broader empirical literature on the determinants of sub-national economic disparities.

Using a panel approach, conditional on region- and country-yearfixed effects, we find no statistically significant evidence that rising regional temperatures are negatively related to regional per capita income. Various robustness checks (e.g., concerning the measurement of regional climatic and economic conditions) support this finding. When we employ a long-difference approach that is more attuned to exploring the long-run relationship between rising temperatures and changes in regional economic activity, we also find no statistically robust evidence that rising temperatures correlate with lower per capita income levels in the long run. To account for potential heterogeneous effects with respect to the temperature-income relationship, we use short- and long-run threshold approaches. We find that country-specific conditions appear to be a key moderator regarding the relationship between temperature and income in the long run. For regions located within countries with weak economic-legal and political institutions (characterized by weak property rights, insufficient access to legal institutions and public goods, weak civil liberties etc.), rising temperatures are negatively associated with regional per capita income in the long run; for regions in countries with strong economic-legal and political institutions, there is no statistically significant association between regional temperature and income. This finding may also speak to the prevalence of intensification effects, implying that the adverse economic impacts of higher temperature may be compounded and become more pronounced over time. We also provide tentative evidence that within vulnerable countries, rising temperatures may constrain long-term regional population and human capital development as potentially important transmission channels from rising temperatures to lower per capita income levels.

Our findings point to a nuanced relationship between regional temperature increases and regional economic development. This may lead to a relevant update of existing priors (Abadie, 2020) concerning the economic consequences of higher temperature which are expected due climate change, while also inviting future research that accounts for the short- and long-run as well as moderating effects of temperature on economic outcomes. Our main dataset starts in 1990. It may be fruitful to move to regional economic data from the 1960s, 1970s and 1980s once such data becomes available, as this may make it more likely to capture the full association of long-run climate change and regional economic development. Our use of data by Gennaioli et al. (2014) starting in 1950 already explores this direction, but this dataset does not cover many regions in Africa for which the economic effects of warming could be more pronounced due to weak economic-legal and political institutions that may make adaptation more difficult. Furthermore, as already discussed above, future research may investigate the role of further economic and politico-institutional factors in moderating the temperature-income relationship at the regional level. For instance, this research may account for the roles of trade or differences in the division of political power between regional and central governments. Finally, when we investigate heterogeneous effects in the temperature-income nexus, we employ variables measuring initial economic, legal and political conditions in the 1980s to ameliorate endogeneity concerns regarding the potential role of climatic conditions in institutional development. Still, one may argue that this does not fully solve the underlying simultaneous equation issue, which consequently invites the use of more elaborate empirical methods such as instrumental-variable threshold models (e.g., Caner and Hansen, 2004).

Global temperature rises linked to climate change are projected to continue for the coming decades (IPCC, 2014, 2021). Our study suggests that rising temperatures are—for the time being—particularly relevant for regions in countries with weak political and economic institutions. On average—and for already richer economies with stronger political and economic institutions—we could detect no direct relationship between temperature and income. For one, this central finding matters for

Appendix A. List of countries

Afghanistan

Albania*

Algeria

Angola

Argentina*

Armenia

Australia

Azerbaijan

Bangladesh*

Austria*

Belarus

Belize

Benin'

Bhutan

Bolivia*

Botswana

Bulgaria*

Burundi

Cambodia

Cameroon

Canada³

Chad

Chile*

China*

Colombia*

Congo (DR)

Costa Rica

Congo (Brazzaville)

Burkina Faso

Brazil*

Bosnia & Herzegovina

Central African Republic

Belgium*

Egypt' El Salvador* Equatorial Guinea Eritrea Estonia' Eswatini Ethiopia Finland* France* Gabon Gambia Georgia Germany* Ghana Greece? Guatemala³ Guinea Guinea Bissau Guyana Haiti Honduras* Hungary India* Indonesia* Iran* Iraq Ireland³ Italv* Jamaica Japan* Jordan* Kazakhstan*

Kenya^{*}

economic models of climate change (see also Tol, 2021). For another, it suggests that the vulnerability to higher temperatures is not constant, which has relevant public policy consequences. For instance, efforts to improve institutional performance at the country-level can help to reduce vulnerability to potential negative regional economic effects of warming. Such efforts can be pursued independent of and in addition to global efforts to reduce greenhouse gas emissions. At the same time, our results are by no means meant to encourage so-called "climate change denial". Global temperatures have increased, and our empirical results point to (heterogeneous) economic losses due to rising temperatures. Finally, we want to emphasize that potential adverse effects of higher temperatures on sub-national per capita income could take longer time horizons to fully materialize and could become more pronounced if future climate change accelerates or leads to very volatile conditions.

CRediT authorship contribution statement

Daniel Meierrieks: Writing – review & editing, Validation, Methodology, Formal analysis, Data curation, Conceptualization. David Stadelmann: Writing – review & editing, Writing – original draft, Validation, Methodology, Formal analysis, Conceptualization.

Declaration of competing interest

None.

Acknowledgements

Libva

Malawi

Mali

Malaysia*

Mauritania

Mexico*

Moldova

Mongolia³

Morocco³

Myanmar

Namibia

Nepal*

Niger

Nigeria*

Norway'

Pakistan*

Palestine

Panama*

Paraguay*

Philippines*

Peru*

Poland*

Portugal*

Romania³

Montenegro

Mozambique*

Netherlands*

New Zealand

North Macedonia*

Papua New Guinea

Nicaragua*

Lithuania*

Madagascar

Funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) under Germany's Excellence Strategy – EXC 2052/1-390713894. The authors have no conflicts of interests to declare.

Slovenia* Somalia South Africa* South Korea³ South Sudan Spain' Sudan Suriname Sweden Switzerland* Svria Tajikistan Tanzania* Thailand' Timor Leste Togo Trinidad & Tobago Tunisia Türkiye Turkmenistan Uganda Ukraine* United Kingdom* United States* Uruguay Uzbekistan* Vanuatu Venezuela* Vietnam* Yemen Zambia Zimbabwe

(continued on next page)

D. Meierrieks and D. Stadelmann

(continued)

Cote d'Ivoire	Kuwait	Russia*	
Croatia*	Kyrgyzstan*	Rwanda	
Cuba	Lao	Saudi Arabia	
Czech Republic*	Latvia*	Senegal	
Denmark*	Lebanon	Serbia*	
Dominican Republic	Lesotho*	Sierra Leone	
Ecuador*	Liberia	Slovakia*	

Note: (*) indicates that this country is also included in the dataset of Gennaioli et al. (2014). This dataset also includes the countries of Sri Lanka and the United Arab Emirates that are not included in the Global Data Lab Dataset.

A.1. Alternative and more complex lag structures

In Table 2, we predict regional per capita income by contemporaneous regional temperature in the previous year (i.e., at *t*). Inspired by Dell et al. (2012), we consider whether alternative and more complex lag structures yield different results in Table A1. For instance, such lag structures may allow us to consider whether potential adverse economic effects of rising temperatures materialize only after some years or whether these unfavorable effects cumulate over time.

We proceed as follows. First, we run models where regional per capita income is explained separately by contemporaneous regional temperature or by regional temperature at *t*-1, *t*-2, *t*-3, *t*-4 and *t*-5, respectively. Second, we allow for cumulative effects, e.g., by predicting regional per capita income by regional temperature at *t*-0 to *t*-5.

As reported in Table A1, all coefficient estimates are never statistically significant both individually and jointly. Thus, in line with our main results reported in the main text, our analysis provides no evidence that increasing temperatures at the regional level are associated with higher or lower per capita income, regardless of which lag structure we employ.

Table A1

Alternative Lag Structures.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
Temperature	0.004						-0.003	-0.001	-0.006	-0.012	-0.010
	(0.009)						(0.009)	(0.011)	(0.010)	(0.012)	(0.011)
Temperature t-1		0.005					-0.000	-0.005	-0.002	-0.003	-0.009
		(0.009)					(0.010)	(0.007)	(0.008)	(0.009)	(0.011)
Temperature t-2			0.000					-0.004	-0.011	-0.007	-0.006
			(0.010)					(0.014)	(0.011)	(0.012)	(0.012)
Temperature t-3				-0.003					-0.007	-0.012	-0.008
				(0.010)					(0.015)	(0.012)	(0.012)
Temperature t-4					-0.006					-0.013	-0.014
					(0.010)					(0.018)	(0.015)
Temperature t-5						-0.013					-0.028
						(0.010)					(0.018)
[Cumulative Effect]							[-0.003]	[-0.010]	[-0.027]	[-0.045]	[-0.076]
[Standard Error]							[0.018]	[0.029]	[0.039]	[0.052]	[0.065]
Region-Fixed Effects	Yes	Yes	Yes	Yes	Yes						
Country*Year-Fixed Effects	Yes	Yes	Yes	Yes	Yes						
Adjusted R ²	0.997	0.997	0.997	0.997	0.997	0.997	0.997	0.997	0.997	0.997	0.997
No. of Observations	40,108	38,807	37,507	36,204	34,907	33,609	38,741	37,411	36,086	34,774	33,466
No. of Regions	1544	1544	1543	1543	1543	1539	1538	1535	1534	1534	1532
No. of Countries	152	152	152	152	152	152	152	152	152	152	151

Notes: Dependent variable (DV) is (logged) regional per capita income. Standard errors clustered at the regional and country-year level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

A.2. Further examination of non-linearities in the temperature-income relationship

In Table 3, we tested for the presence of a non-linear relationship between regional temperature and income by considering a 17 °C threshold. We found no evidence in favor of a non-linear relationship between the two variables. As a robustness check, we explore potential non-linearities by considering various further temperature thresholds. Here, we follow other results from the empirical literature (e.g., Nordhaus, 2006; Deryugina and Hsiang, 2014; Burke et al., 2015a, 2018; Zhao et al., 2018; Li et al., 2019) and consider temperature thresholds at 9, 13, 21 and 24 °C, respectively.

As reported in Table A2, higher temperatures are not associated with economic activity below and above the various temperature thresholds in statistically meaningful ways. That is, in line with our main results, there is no clear evidence for non-linear links between increasing temperatures and per capita income at the regional level.

Table A2

Alternative Temperature Thresholds.

	(1)	(2)	(3)	(7)
Temperature Threshold \rightarrow	9 °C	13 °C	21 °C	24 °C
Temperature (<threshold)< td=""><td>0.007</td><td>0.006</td><td>0.004</td><td>0.005</td></threshold)<>	0.007	0.006	0.004	0.005
	(0.008)	(0.008)	(0.009)	(0.009)
Temperature (>Threshold)	-0.083*	-0.105	-0.008	-0.248
	(0.047)	(0.086)	(0.194)	(0.212)
			(contin	ued on next page)

Table A2 (continued)

	(1)	(2)	(3)	(7)
Region-Fixed Effects	Yes	Yes	Yes	Yes
Country*Year-Fixed Effects	Yes	Yes	Yes	Yes
Adjusted R ²	0.997	0.997	0.997	0.997
No. of Observations	40,108	40,108	40,108	40,108
No. of Regions	1544	1544	1544	1544
No. of Countries	152	152	152	152

Notes: Dependent variable (DV) is (logged) regional per capita income as described in the text. Standard errors clustered at the regional and country-year level in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

A.3. Further sources of country-level heterogeneity

For our baseline approach, we consider a variety of economic, legal and politico-institutional variables as potential threshold variables, i.e., as moderators accounting for potential heterogeneity in the temperature-income relationship in the short as well as long run. Clearly, there are many additional variables that may likewise moderate the relationship between regional temperature and income. As an empirical extension, we focus on four additional sources of heterogeneity: a country's dependence on agricultural and industrial production, respectively, the country-level exposure to temperature changes as well as the economic response to the shocks of warming at the country level.

First, to study the role of a country's agricultural sector, we collect the country-specific *value added by agriculture as a share of GDP* as a moderator from the *World Development Indicators*. For instance, higher temperatures might depress economic output by hurting agricultural production. This could imply that a stronger dependence on agriculture might disproportionately increase regional economic vulnerability to rising temperatures.

Second, to explore the role of a country's level of industrial development in the regional temperature-income nexus, from the *World Development Indicators* we draw the *value added by manufacturing as a share of GDP*. For example, a larger industrial sector ought to imply a smaller agricultural sector, potentially meaning that more industrialized countries are less vulnerable to climate shocks. Alternatively, the adverse relationship between rising temperatures and labor productivity may make regions in industrialized countries (which rely on high levels of labor productivity) also vulnerable to some extent, e.g., when cooling technologies are not employed.

Third, it is also possible that the local impact of warming on regional economic activity is different depending on how strongly countries (as a whole) are subject to warming. For instance, when a country as a whole does not experience strong warming, policy measures to foster adaption would be less likely to be at the top of the central government's political agenda. This, in turn, could imply that regions within such countries that do experience warming may receive fewer resources from the central government and thus be more susceptible to the ill local economic effects of warming. Below, we operationalize a country's susceptibly to warming (as a moderator variable) as the *country-level change in temperature between 1990–1993 and 2014–2017*. In other words, we use the distribution of temperature change between 1990 and 2017 to create a moderator variable to distinguish between countries with different levels of warming.

Finally, regional economic activity may respond differently to local warming depending on whether a region is located within a country that has a high or low aggregate (i.e., country-specific) economic response to temperature shocks. For example, if there were economic losses due to warming at the country level, this may mean that there are fewer resources available to counter the adverse effects of warming at the local level. To study the role of aggregate economic effects of higher temperatures on the regional economy, we proceed as follows. First, we estimate the relationship between income and temperature at the country level separately for each country in the sample. Second, we estimate how temperature affects income for each country regions, using the *associated point estimates for each country* (i.e., the aggregate economy) as a moderator variable to split the dataset into countries with low and high responses (in terms of the size of the point estimates) to temperature increases.

We report our findings in Table A3. Keeping in mind that sample sizes differ to data availability and using our usual short- and long-run threshold approaches, we find that in the panel setting, the link between temperature and regional economic activity does not differ for regions located in countries with low versus high levels of agricultural or industrial development. They also do not differ for regions located in countries that experience low versus high aggregate temperature increases or that have a low versus high correlation between temperature and income at the country level (aggregate economy). Furthermore, we find that in the long-difference setting, there is some (statistically rather weak) evidence that higher temperatures are negatively associated with regional economic development when manufacturing (but not agriculture) plays no strong role in the country of interest. Finally, in the long-difference setting, we also find that country-level differences in the extent of temperature changes matter, while weaker or stronger associations between temperature and the aggregate economy play no statistically significant role for the link between regional temperature and income. For the extent of temperature changes, we find that regions that experience higher temperature increases but that are located in countries that have been warming less than others (where, i.e., the associated Moderator = 0), show a stronger negative association between the temperature increase and regional income than regions in countries have been warming more strongly. This may indicate that regions in countries experiencing lower temperature increases receive fewer resources from the central government to adapt to rising temperature because the central government feels less political pressure or has fewer incentives to foster adaption. Furthermore, economic agents may relocate (e.g., via internal migration or business relocation) in such situations to other parts of a country when specific regions are strongly affected by higher temperature as warming is more moderate in other parts of the country. As such, these heterogeneous relationships in the long run point to potential future research directions when exploring potential effects of higher temperature. Here, adaptation possibilities need not only be moderated by (poor) legal and economic institutions but potentially also by existing experience with recent increases in temperature.

In general, the findings reported in Table A3 are in line with our main findings in that (1) temperature increases may only matter to regional income in the long run and (2) in a heterogeneous manner, where the adverse effects of warming are especially felt in regions that are plausibly more vulnerable or have less experience with potential adaptation possibilities.

Table A3

Role of Further Country-Level Moderators.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Empirical Approach \rightarrow	Panel Approach				Long-Difference Approach				
Moderator →	Agriculture	Manufacturing	Temperature Change	Aggregate Economy	Agriculture	Manufacturing	Temperature Change	Aggregate Economy	
Temperature (Moderator =									
0)	-0.099	-0.002	-0.135	-0.013					
	(0.087)	(0.033)	(0.108)	(0.029)					
Temperature (Moderator =									
1)	0.016	-0.010	0.009	-0.030					
	(0.031)	(0.093)	(0.019)	(0.033)					
Δ Temperature (Moderator									
= 0)					-0.083	-1.249*	-1.369***	-0.517	
					(0.122)	(0.657)	(0.385)	(0.352)	
Δ Temperature (Moderator									
= 1)					-0.629	-0.136	0.088	0.063	
					(0.395)	(0.133)	(0.093)	(0.099)	
[Equality of Coefficients Test									
p-value]	[0.21]	[0.33]	[1.75]	[0.16]	[0.19]	[0.09]*	[13.50]***	[0.11]	
Threshold Estimate	10.02	15.81	0.04 °C	0.03	20.56	7.76	0.04 °C	-0.50	
Region-Fixed Effects	Yes	Yes	Yes	Yes					
Country*Period-Fixed									
Effects	Yes	Yes	Yes	Yes					
Country Dummies					Yes	Yes	Yes	Yes	
Period Dummy (Intercept)					Yes	Yes	Yes	Yes	
Adjusted R ²	0.994	0.995	0.996	0.996	0.993	0.994	0.996	0.996	
No. of Observations	24,192	21,840	35,700	35,700	873	787	1289	1289	
No. of Regions	864	780	1275	1275	873	787	1289	1289	
No. of Countries	80	72	123	123	82	73	125	125	

Notes: Dependent variable (DV) is (logged) per capita income in Models (1) to (4) and Δ (logged) per capita income in Models (5) to (8). Δ always refers to the difference between the early and late period (1990–1993 vs. 2014–2017). "Agriculture" is the value added (as a share of GDP) from agriculture, forestry and fishing. "Manufacturing" is the value added (as a share of GDP) from manufacturing. "Temperature Change" is the change in temperature between 1990 and 1993 and 2014–2017 at the country level. "Aggregate Economy" is the point estimate associated with the individual country-level effect of temperature on country-level per capita income. The threshold estimates refer to the specific moderator variable shares. For the moderator variable, "Moderator = 1" refers to a relatively large agricultural or manufacturing sector in the country of interest, respectively. Standard errors clustered at the regional and country-year level in parentheses for Models (1) to (4). Robust standard errors in parentheses for Models (5) to (8). *** p < 0.01, ** p < 0.05, * p < 0.1.

Appendix B. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.eneco.2024.107758.

References

- Abadie, Alberto, 2020. Statistical non-significance in empirical economics. AER: Insights 2 (2), 193–208.
- Acemoglu, Daron, Johnson, Simon, 2005. Unbundling Institutions. J. Polit. Econ. 113 (5), 949–995.
- Auffhammer, Maximilian, Hsiang, Solomon M., Schlenker, Wolfram, Sobel, Adam, 2013. Using weather data and climate model output in economic analyses of climate change. Rev. Environ. Econ. Policy 7 (2), 181–198.
- Barreca, Alan, 2012. Climate change, humidity, and mortality in the United States. J. Environ. Econ. Manag. 63 (1), 19–34.
- Barreca, Alan, Deschênes, Olivier, Guld, Melanie, 2018. Maybe next month? Temperature shocks and dynamic adjustments in birth rates. Demography 55 (4), 1269–1293.
- Beine, Michel, Parsons, Christopher, 2015. Climatic factors as determinants of international migration. Scand. J. Econ. 117 (2), 723–767.
- Berlemann, Michael, Steinhardt, Max F., 2017. Climate change, natural disasters, and migration: a survey of the empirical evidence. CESifo Econ. Stud. 63 (4), 353–385.
- Bolt, Jutta, van Zanden, Jan L., 2024. Maddison-style estimates of the evolution of the world economy: a new 2023 update. J. Econ. Surveys. https://doi.org/10.1111/ joes.12618.
- Brückner, Markus, Ciccone, Antonio, 2011. Rain and the democratic window of opportunity. Econometrica 79 (3), 923–947.
- Burke, Marshall, Emerick, Kyle, 2016. Adaptation to climate change: Evidence from US agriculture. Am. Econ. J. Econ. Pol. 8 (3), 106–140.
- Burke, Marshall, Hsiang, Solomon M., Miguel, Edward, 2015a. Global non-linear effect of temperature on economic production. Nature 527 (7577), 235–239.
- Burke, Marshall, Hsiang, Solomon M., Miguel, Edward, 2015b. Climate and conflict. Ann. Rev. Econ. 7, 577–617.
- Burke, Marshall, Matthew Davis, W., Diffenbaugh, Noah S., 2018. Large potential reduction in economic damages under UN mitigation targets. Nature 557 (7706), 549–553.

- Cameroon, A. Colin, Gelbach, Jonah B., Miller, Douglas L., 2011. Robust inference with multiway clustering. J. Bus. Econ. Stat. 29 (2), 238–249.
- Caner, Mehmet, Hansen, Bruce H., 2004. Instrumental variable estimation of a threshold model. Economet. Theor. 20, 813–843.
- Carleton, Tamma A., Hsiang, Solomon M., 2016. Social and economic impacts of climate. Science 353 (6304), aad9837.
- Carter, Colin, Cui, Xiaomeng, Ghanem, Dalia, Mérel, Pierre, 2018. Identifying the economic impacts of climate change on agriculture. Ann. Rev. Resour. Econ. 10 (1), 361–380.
- Cattaneo, Cristina, Peri, Giovanni, 2016. The migration response to increasing temperatures. J. Dev. Econ. 122 (3), 127–146.

Choi, In, 2001. Unit root tests for panel data. J. Int. Money Financ. 20 (2), 249–272. Christensen, Peter, Gillingham, Kenneth, Nordhaus, William, 2018. Uncertainty in

- forecasts of long-run economic growth. Proc. Natl. Acad. Sci. 115 (21), 5409–5414. Colacito, Riccardo, Hoffmann, Bridget, Phan, Toan, 2019. Temperature and growth: a
- panel analysis of the United States. J. Money Credit Bank. 51 (2–3), 313–368.
- Coppedge, Michael, Gerring, John, Knutsen, Carl Henrik, Lindberg, Staffan I., Teorell, Jan, Altman, David, Bernhard, Michael, Cornell, Agnes, Fish, M. Steven, Gastaldi, Lisa, Gjerløw, Haakon, Glynn, Adam, Hicken, Allen, Lührmann, Anna, Maerz, Seraphine F., Marquardt, Kyle L., McMann, Kelly, Mechkova, Valeriya, Paxton, Pamela, Pemstein, Daniel, von Römer, Johannes, Seim, Brigitte, Sigman, Rachel, Skaaning, Svend-Erik, Staton, Jeffrey, Sundtröm, Aksel, Tzelgov, Eitan, Uberti, Luca, Wang, Yi-ting, Wig, Tore, Ziblatt, Daniel, 2021. V-Dem Dataset V11.1. In: Varieties of Democracy (V-Dem) Project. Available at https ://www.v-dem.net/en/data/data/v-dem-dataset-v111/ (accessed May 1, 2024).
- Cuaresma, Crespo, Jesús, Gernot Doppelhofer, Feldkircher, Martin, 2014. The determinants of economic growth in European regions. Reg. Stud. 48 (1), 44–67.
- Deaton, Angus, 2010. Instruments, randomization, and learning about development. J. Econ. Lit. 48 (2), 424–455.
- Deaton, Angus, 2013. The Great Escape. Princeton University Press, Princeton.
- Dell, Melissa, Jones, Benjamin F., Olken, Benjamin A., 2009. Temperature and income: reconciling new cross-sectional and panel estimates. Am. Econ. Rev. 99 (2), 198–204.

D. Meierrieks and D. Stadelmann

- Dell, Melissa, Jones, Benjamin F., Olken, Benjamin A., 2012. Temperature shocks and economic growth: evidence from the last half century. Am. Econ. J. Macroecon. 4 (3), 66–95.
- Dell, Melissa, Jones, Benjamin F., Olken, Benjamin A., 2014. What do we learn from the weather? The new climate–economy literature. J. Econ. Lit. 52 (3), 740–798.

Demetrescu, Matei, Hanck, Christoph, 2012. A simple nonstationary-volatility robust panel unit root test. Econ. Lett. 117 (1), 10–13.

- Deryugina, Tatyana, Hsiang, Solomon M., 2014. Does the environment still matter? Daily temperature and income in the United States. In: Working Paper 20750. National Bureau of Economic Research.
- Deryugina, Tatyana, Hsiang, Solomon M., 2017. The Marginal Product of Climate. Working Paper 24072.. National Bureau of Economic Research.
- Deschênes, Olivier, 2014. Temperature, human health, and adaptation: a review of the empirical literature. Energy Econ. 46 (3), 606–619.
- Deschênes, Olivier, Greenstone, Michael, 2007. The economic impacts of climate change: evidence from agricultural output and random fluctuations in weather. Am. Econ. Rev. 97 (1), 354–385.
- Gallup, John Luke, Sachs, Jeffrey D., Mellinger, Andrew D., 1999. Geography and economic development. Int. Reg. Sci. Rev. 22 (2), 179–232.
- Gennaioli, Nicola, La Porta, Rafael, De Silanes, Florencio Lopez, Shleifer, Andrei, 2013. Human capital and regional development. Q. J. Econ. 128 (1), 105–164.
- Gennaioli, Nicola, La Porta, Rafael, De Silanes, Florencio Lopez, Shleifer, Andrei, 2014. Growth in regions. J. Econ. Growth 19 (3), 259–309.
- Greßer, Christina, Meierrieks, Daniel, Stadelmann, David, 2021. The link between regional temperature and regional incomes: econometric evidence with sub-National Data. Econ. Policy 36 (107), 523–550.
- Hansen, Bruce E., 1999. Threshold effects in non-dynamic panels: estimation, testing, and inference. J. Econ. 93 (2), 345–368.
- Helbling, Marc, Meierrieks, Daniel, 2021. How climate change leads to emigration: conditional and long-run effects. Rev. Dev. Econ. 25 (4), 2323-2349.
- Herwartz, Helmut, Siedenburg, Florian, 2008. Homogenous panel unit root tests under cross sectional dependence: finite sample modifications and the wild bootstrap. Comp. Stat. Data Anal. 53 (1), 137–150.
- Hsiang, Solomon M., 2010a. Temperatures and cyclones strongly associated with economic production in the Caribbean and Central America. Proc. Natl. Acad. Sci. 107 (35), 15367–15372.
- Hsiang, Solomon M., 2010b. Climate Econometrics. Ann. Rev. Resour. Econ. 8 (1), 43–75.
- Intergovernmental Panel on Climate Change [IPCC], 2014. Climate change 2014: Synthesis report. In: Team, Core Writing, Pachauri, R.K., Meyer, L.A. (Eds.), Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. IPCC, Geneva, Switzerland. Available at https://www.ipcc.ch/report/ar5/syr/ (accessed May 1, 2024).
- Intergovernmental Panel on Climate Change [IPCC], 2021. Summary for policymakers. In: Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S.L., Péan, C., Berger, S., Caud, N., Chen, Y., Goldfarb, L., Gomis, M.I., Huang, M., Leitzell, K., Lonnoy, E., Matthews, J.B.R., Maycock, T.K., Waterfield, T., Yelekci, O., Yu, R., Zhou, B. (Eds.), Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press. In Press. Available at https://www.ipcc.ch/report/ sixth-assessment-report-working-group-i/ (accessed May 1, 2024).
- Jetter, Michael, Mösle, Saskia, Stadelmann, David, 2019. Cursed by no coast: how regional Landlockedness affects income within countries. Econ. Lett. 181, 70–73.
- Johnson, Simon, Larson, William, Papageorgiou, Chris, Subramanian, Arvind, 2013. Is newer better? Penn world table revisions and their impact on growth estimates. J. Monet. Econ. 60 (2), 255–274.
- Jolliffe, Ian T., 2002. Principal Component Analysis. Springer, New York.
- Kahn, Matthew E., Mohaddes, Kamiar, Ryan, N.C., Ng, M. Hashem, Pesaran, Mehdi Raissi, Yang, Jui-Chung, 2021. Long-term macroeconomic effects of climate change: a cross-country analysis. Energy Econ. 104 (12), 105624.
- Kalkuhl, Matthias, Wenz, Leonie, 2020. The impact of climate conditions on economic production: evidence from a global panel of regions. J. Environ. Econ. 103 (3), 102360.

- Kotz, Maximilian, Levermann, Anders, Wenz, Leonie, 2022. The effect of rainfall changes on economic production. Nature 601, 223–227.
- Kotz, Maximilian, Levermann, Anders, Wenz, Leonie, 2024. The economic commitment of climate change. Nature 628, 551–557.
- Lam, David A., Miron, Jeffrey A., 1996. The effects of temperature on human fertility. Demography 33, 291–305.
- Lanzafame, Matteo, 2014. Temperature, rainfall and economic growth in Africa. Empir. Econ. 46 (1), 1–18.
- Lemoine, Derek, 2021. Estimating the Consequences of Climate Change from Variation in Weather. Working Paper 25008. National Bureau of Economic Research.
- Letta, Marco, Tol, Richard S.J., 2019. Weather, climate and Total factor productivity. Environ. Resour. Econ. 73, 283–305.
- Li, Ning, Bai, Kou, Zhang, Zhengtao, Feng, Jieling, Chen, Xi, Liu, Li, 2019. The nonlinear relationship between temperature changes and economic development for individual provinces in China. Theor. Appl. Climatol. 137 (3), 2477–2486.
- Meierrieks, Daniel, 2021. Weather shocks, climate change and human health. World Dev. 138, 105228.
- Miguel, Edward, Satyanath, Shanker, Sergenti, Ernest, 2004. Economic shocks and civil conflict: an instrumental variables approach. J. Polit. Econ. 112 (4), 725–753.
- Mitton, Todd, 2016. The wealth of subnations: geography, institutions, and withincountry development. J. Dev. Econ. 118 (1), 88–111.
- Mohaddes, Kamiar, Ryan, N.C., Ng, M. Hashem, Pesaran, Mehdi Raissi, Yang, Jui-Chung, 2023. Climate change and economic activity: evidence from US states. Oxford Open Econ. 2, odac010.
- Nath, Ishan B., 2020. The Food Problem and the Aggregate Productivity Consequences of Climate Change. Working Paper 27297, National Bureau of Economic Research.
- Nordhaus, William D., 2006. Geography and macroeconomics: new data and new findings. Proc. Natl. Acad. Sci. USA 103 (10), 3510–3517.
- North, Douglass C., 1981. Structure and Change in Economic History. Norton, New York. Park, R. Jisung, 2022. Hot temperature and high-stakes performance. J. Hum. Resour. 57 (2), 400–434.
- Permanyer, Iñaki, Smits, Jeroen, 2020. Inequality in human development across the globe. Popul. Dev. Rev. 46 (3), 583–601.
- Schlenker, Wolfram, Lobell, David B., 2010. Robust negative impacts of climate change on African agriculture. Environ. Res. Lett. 5, 014010.
- Sellers, Samuel, Gray, Clark, 2019. Climate shocks constrain human fertility in Indonesia. World Dev. 117 (5), 357–369.
- Smits, Jeroen, 2016. Sub-National Development Indicators for Research and Policy-Making. GDL Working paper 16–101.
- Smits, Jeroen, Permanyer, Iñaki, 2019. The subnational human development database. Sci. Data 6, 190038.
- Tol, Richard S.J., 2009. The economic effects of climate change. J. Econ. Perspect. 23 (2), 29–51.
- Tol, Richard S.J., 2021. Do climate dynamics matter for economics? Nat. Clim. Chang. 11, 802–803.
- Weil, David N., 2013. Economic Growth. Pearson, Harlow.
- Willmott, Cort J., Matsuura, Kenji, 2001. Terrestrial Air Temperature and Precipitation: Monthly and Annual Time Series (1950–1999). Available at http://climate.geog. udel.edu/~climate/html_pages/README.ghcn_ts2.html (accessed May 1, 2024).
- World Bank, 2021. World Development Indicators. World Bank, Washington, D.C. Zhao, Xiaobing, Gerety, Mason, Kuminoff, Nicolai V., 2018. Revisiting the temperature-
- economic growth relationship using global subnational data. J. Environ. Manag. 223, 537–544.
- Zhao, Qi, et al., 2021. Global, regional, and National Burden of mortality associated with non-optimal ambient temperatures from 2000 to 2019: a three-stage modelling study. Lancet Planet. Health 5 (7), e415–e425.
- Zivin, Joshua G., Shrader, Jeffrey, 2016. Temperature extremes, health, and human capital. Futur. Child. 26 (1), 31–50.
- Zivin, Joshua G., Hsiang, Solomon M., Neidell, Matthew, 2018. Temperature and human Capital in the Short and Long run. J. Assoc. Environ. Resour. Econ. 5 (1), 77–105.